Author's response to reviews

Title: Lessons learned from implementation of a demonstration program to reduce the burden of anaemia and hookworm in women in Yen Bai Province, Viet Nam

Authors:

Tran Q Phuc (tquangphuc@yahoo.com)
Seema Mihrshahi (seemam@unimelb.edu.au)
Gerard J Casey (gcasey@unimelb.edu.au)
Luong B Phu (phuytyb@yahoo.com.vn)
Nong T Tien (nongthitien@yahoo.com)
Sonia R Caruana (sonia2rose@yahoo.com.au)
Tran D Thach (indthach@yahoo.com)
Antonio Montresor (montresora@vtn.wpro.who.int)
Beverley-Ann Biggs (babiggs@unimelb.edu.au)

Version: 6 Date: 13 July 2009

Author's response to reviews:

Dear Editorial Production Team,

The proof read and corrected manuscript and Figure 1 have now been uploaded. Please refer the email from Gerard Casey (10/07/09) re the technical problem we had on Friday. These uploaded files are the final version.

Many thanks for your assistance.

Regards

B-A Biggs